
Pager Report 
5/24 - 6/6, Minsuk/Koji/Rolf 

�

Weekly News What happened Related Runs 
(Beam status) 

Fri 5/24, 10:17 
Fri 5/24, 12:00 

PSM alarm, all east channels empty in YMon 
⇒  Koji accessed with Rob, Oh dear! Remained 
power cable unplugged (somebody forgot to plug it 
back in this morning access) 

(from yesterday 
access, all cosmic) 
145230, 145231, 
145232 

Sat 5/25, 14:40 PC froze ⇒  Ace Gervasio rebooted PC (no beam) 

Sun 5/26, 04:06 
 
Sun 5/28, 05:39 

TOF trigger inhibit and paged →  YMon okay 
⇒  PC didn’t properly work (SMACS/iFix okay) 
Site-wide power outage, Emergency! 
⇒  Recovering whole day 

145299, 145300, 
145301 
(all files no exist) 

   

Mon 5/27, 21:00 
 
 
Sun 5/28, 23:30 
Tue 5/28, 01:30 

PSM alarm, b0wcal05 failed and TOF inhibit 
⇒  PSM masked and TOF inhibit disabled, 

while CO Stefano tried to figure it out 
He paged me and I asked to page ADMEM expert 
⇒  Reset wcal05, which have had troubles 

(low Lum <10 1nb− )
 
 
 
good run but no file 

Wed 5/29, 13:40 
~ Thu 5/30, 16:00 

Tevatron study  

   

Fri 5/31, 00:39 
Fri 5/31, 01:00 
 
Fri 5/31, 06:00 

Beam stored with low luminosity 
Heartbeat lost, SMACS frozen (but PC fine) 
⇒  Ace Gervasio restarted SMACS within 1 min 
Heartbeat lost again. He forced system off 

(L2 problem/test) 

Sat 6/1, 09:00 While stopping run, iFix problem 
⇒  SciCo Sham forced to turn off HV and paged 

Good run! 
 

Sun 6/2, 15:00  
 
Sun 6/2, 17:30 

Beam lost because of upstream accelerator problem 
⇒  No more store 
Magnet off, TOF system shutdown for outage 
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Shutdown Plan: about 2 weeks, Jun 3 - 14 (or 16, 17) 
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Run summary from ‘End of Shift Report’ with default physics table 
Mon May 27 Tue May 28 Fri May 31 Sat Jun 1 Sun Jun 2 
145378 (4) 145411 (40) 

145414 (12) 
145415 (14) 
145418 (13) 
145420 (55) 
145427 (9) 
145430 (9) 
145450 (27) 

145532 (53) 
145602 (6) 
145608 (169) 
145613 (29) 

145650 (19) 
145651 (23) 
145653 (46) 
145654 (117) 

145669 (17) 

1(4 )nb−  1(179 )nb−  1(257 )nb−  1(205 )nb−  1(17 )nb−  

 
Fig 1. East Power Supply unplugged�

 
Fig 2. PS alarm due that wcal05 lost +5V 
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